\documentclass[12pt,twoside,a4paper]{article}
\usepackage[english]{babel}
\usepackage{pdfsync}
\newcommand{\thepages}{1--4}
\newcommand{\doinumber}{001}

\usepackage{umj_8_1}            % Style package of Ural Mathematical Journal

\usepackage{amsmath}            % If the package features are used
\usepackage{amsfonts,amssymb}   % If there is need to use math alphabets
                                % \mathbb{} - amsfonts.sty
                                % \mathfrak{} - eufrak.sty, amsfonts.sty
                                % symbols of font Euler - amssymb.sty
\newcommand{\Leftclt}{David Aron and Santosh Kumar}   % Left page header - name & family name of article author
\newcommand{\Rightclt}{Fixed point theorem for multivalued non-self mappings satisfying JS-contraction...} %% Right page header - Short title of article
\makeatletter\def\Hy@xspace@end{}\makeatother



%\usepackage{refcheck}
%\setcounter{page}{13} %��������� � ...
\sloppy
\usepackage{srcltx}             % For easy move between WiNEdT and YAP

\begin{document}
\begin{thebibliography}{99}
	
	\bibitem{Alghamdi}{Alghamdi, Maryam A., Berinde,Vasile, Shahzad, Naseer, Fixed Points of Multivalued Nonself AlmostContractions, \emph{Journal of Applied Mathematics}, vol. 2013, Article ID 621614, 6 pages,2013. https://doi.org/10.1155/2013/621614}
	
	\bibitem{Altun}{Altun, I., and Minak, G., An extension of Assad-Kirk?s fixed point theorem for multivalued nonself mappings, \emph{Carpathian. J. Math.}, 2016, Vol. 32, 147--155.}
	
	\bibitem{Aron}{Aron, D. and Kumar, S., Fixed point theorem for a sequence of multivalued nonself mappings in metrically convex metric spaces, \emph{Topological Algebra and Its Applications}, 10, (2022), 1-12.} 
		
	\bibitem{Assad}{Assad, N. A.  and Kirk, W. A.,  Fixed point theorems for multivalued mappings of contractive type, \emph{Pacific. J. Math.}, 1972, Vol. 43, 553--562.}
	
	\bibitem{Damjanovic}{Damjanovic, B., Samet, B. and Vetro, C., Common fixed point theorems for multi-valued maps, \emph{Acta Math. Sci. (English Ed.)} 2012, Vol. 32, 818?824.}
	
	\bibitem{Hussain}{Hussain, N., Parvaneh, V., Samet, B. and Vetro, C., Some fixed point theorems for generalized contractive mappings in complete metric spaces, \emph{Fixed Point Theory Appl.}, 2015, 185.}
	
	\bibitem{Imdad}{Imdad, M. and  Kumar, S., Rhoades type fixed point theorems for a pair of non-self mappings, \emph{J. computers Math. Appl.}, 46(2003), 919-927.} 
		
	\bibitem{Itoh}{Itoh, S., Multivalued generalized contractions and fixed point theorems, Comment, \emph{Math. Univ. Carolinae}, 1977, Vol. 18,  247-?258.}
	
	\bibitem{Jleli}{Jleli, M. and Samet, B., A new generalization of the Banach contraction principle, \emph{J. Inequal. Appl.},2014, Vol. 38, 8 pp.}
	
	\bibitem{Kreyszig}{Kreyszig,E., Introductory functional analysis with applications, \emph{John Wily \& Sons.Inc.}, 1978}
	
	\bibitem{Kumar}{Kumar, S., Rugumisa, T. and Imdad, M., Common Fixed Points in Metrically Convex Partial Metric Spaces, \emph{Konuralp Journal of Mathematics}, Vol. 5 (2)(2017), 56-71.}
		
	\bibitem{Maleknejad}{Maleknejad, K., and Torabi, P., Application of fixed point method for solving nonlinear volterra-hammerstein integral equation, \emph{U.P.B. Sci. Bull Series A}, 2012, 74.}
	
	\bibitem{Nadler}{Nadler, S. B.,  Multivalued contraction mappings,\emph{Pacific. J. Math.}, 1969, Vol. 30, 475--488.}
	
\end{thebibliography}


\date{}

\Mainclt            % Page header print on the first page
% Enter article title and authors
% Reference to grants is a footnote to article title. You may use also Acknowledgments section for this purpose if you wish.

\begin{Titul}
{\large \bf \uppercase{Fixed point theorem for multivalued non-self mappings satisfying JS-contraction with an application}}
%\footnote{This work was supported by ... (project No. NN-N). You may use also Acknowledgments section for this purpose if you wish.}
\\[3ex]
{{\bf David Aron} \\[2ex] {\small Department of Mathematics, College of Natural and Applied Sciences,\\ University of Dar es Salaam, Tanzania, \\ \hrefemail{d.aron.da6@gmail.com} }\\[3ex]
{\bf Santosh Kumar} \\[2ex] {\small Department of Mathematics, College of Natural and Applied Sciences,\\ University of Dar es Salaam, Tanzania\\ \hrefemail{drsengar2002@gmail.com} }\\[5ex]}
\end{Titul}


\begin{Annot}
{\bf Abstract:} In this paper, we present some fixed point results for multivalued non-self mappings. We generalize the fixed point theorem due to Altun and Minak \cite{Altun} by using Jleli and Sameti \cite{Jleli} $\vartheta$-contraction. To validate the results proved here, we provide our main result an appropriate application.

{\bf Keywords:} JS-contraction mapping, multivalued mapping, metric space, non-self mapping, fixed point.

 \textbf{Mathematics Subject Classification 2010:}   47H10,  54H25.\\
\end{Annot}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
 %Please replace what follows by the body %%
 %of your article                         %%
                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction and Preliminaries} %This section must have a number.
\setcounter{equation}{0}
In 1922, in Banach's PhD thesis a remarkable fixed point theorem well known as the Banach contraction principal was initiated. It's simplicity, usefulness and application made it a supreme tool in finding the existence and uniqueness of solution in numerous branches of mathematical analysis and applied sciences. Following the Banach contraction principal, some authors, Nadler \cite{Nadler}, Assad and Kirk \cite{Assad}, Itoh \cite{Itoh} and several others have extended and generalized this theorem in several ways. In fact, Nadler \cite{Nadler} introduced the concept of using Hausdorff metric on multi-valued contraction of self mappings in the study of fixed points. Assad and Kirk \cite{Assad} proved the Banach contraction mapping theorem for multi-valued contraction of non-self mappings and Itoh \cite{Itoh} generalized the theorems due to Assad and Kirk, and many other researchers have made significant contributions in this area (see \cite{Aron,Imdad,Kumar} ). In 2013, Alghamdi \emph{et al.} \cite{Alghamdi} proved fixed point results for multivalued nonself almost contractions on convex metric spaces. Recently, Altun and Minak \cite{Altun} introduced a new approach to Assad and Kirk fixed point theorem and a new real generalization of it, by using $\vartheta-$ contractiveness of a multivalued mapping. Jleli and Samet \cite{Jleli} introduced  $\vartheta-$ contraction and established a new fixed point theorem for such mappings in the setting of generalized metric spaces. Following the notion $\vartheta$, Hussain \emph{et al.} \cite{Hussain} supposed that $\varTheta$ is the set of all functions $\vartheta : [0,\infty)\rightarrow [1,\infty)$ satisfying the following conditions:

$(\vartheta_{1})$ $\vartheta$ is nondecreasing and $\vartheta (t)=1$ if and only if $t=0$;

$(\vartheta_{2})$ for each sequence $\{t_{n}\}\subseteq (0,\infty)$, $\lim \limits_{n\to\infty} \vartheta (t_{n})=1$ if and only if the limit of $\lim \limits_{n\to\infty} t_{n}=0$;

$(\vartheta_{3})$ there exists $r\in (0,1)$ and $l\in (0,\infty]$ such that $\lim \limits_{t\to 0^{+}} \dfrac{\vartheta(t)-1}{t^{r}}=l$;

$(\vartheta_{4})$ $\vartheta (a+b)\leq \vartheta (a) \vartheta (b)$ for all $a,b>0$.  

Throughout this paper we shall denote by $\varTheta$ the set of all functions $\vartheta$ satisfying $(\vartheta_{1})-(\vartheta_{4})$.  

Next, we present some definitions and preliminaries that are required to prove the main result of this paper.

Since we are dealing with multivalued mapping it is important to state a brief description of the Hausdorff metric.\\ The Hausdorff metric measures the distance between subsets of a metric space.
One among many interesting properties of this metric space, which
will be our focus in this paper is that the Hausdorff induced
metric space is complete if our original metric space is complete. Now, we define the Hausdorff metric as follows: 
\begin{dfnen}\emph{\cite{Damjanovic}}
	Let $(M,\varrho)$ be a metric space. Denote by $CB(M)$ the collection of non-empty closed bounded subsets of $M$. For $A,B\in CB(M)$ and $u\in M$, define
	$$\rho(u,A)=\inf_{a\in A} \varrho(u,a)$$ and
	$$\mathcal{H}(A,B)=\max \Big\{\sup_{a\in A}\rho(a,B),\sup_{b\in B}\rho(b,A)\Big\}.$$
	It is seen that $H$ is a metric on $CB(M)$. $H$ is called the Hausdorff metric induced by $\varrho$. The completion of  $(M,\varrho)$ implies that $(CB(M),H)$ is a complete metric space.
\end{dfnen}
Before proceeding further, it is important to know that we will need an extra condition, call it $(\vartheta_{5})$, a very useful part of our tool to help us prove our main results in multivalued mapping and $\varTheta$ satisfy $(\vartheta_{5})$.

$(\vartheta_{5})$ $\vartheta (inf A) = inf \vartheta (A)$ for all $A\subset (0,\infty)$ with inf $A>0$.

The following definition is important for future work in this paper.
\begin{dfnen}\emph{\cite{Kreyszig}.}\\
	(i) A sequence $\{u_n\}$ in a metric space  $(M, \varrho)$ is said to converge or to be convergent if there is a $u \in M$ such that  $$\lim_{n\to \infty} \varrho(u_{n},u)=0.$$
	
	(ii) A sequence $\{u_n\}$ in a metric space $(M, \varrho)$ is said to be  Cauchy sequence if for every $\epsilon >0$ there is an $N=N(\epsilon)$ such that
	$$\varrho(u_{n},u_{m})<\epsilon$$ for every $m,n>N$.
	
	(iii) A metric space $(M, \varrho)$ is said to be complete if every Cauchy sequence in $M$ converges to an element of $M$.
\end{dfnen}  

The following is the description of a metrically convex metric space and some of its properties are stated.

The following definition is due to Assad and Kirk \cite{Assad}.

\begin{dfnen}\emph{\cite{Assad}}
	A metric space $(M,\varrho)$ is said to be metrically convex if for any $u,v\in M$ with $u\neq v$, there exists a point $z\in M$,$(u\neq z\neq v)$ such that 
	\[\varrho(u,v)=\varrho(u,z)+\varrho(z,v).\]
\end{dfnen}
The following result is taken from Assad \cite{Assad} where $\partial K$ denotes the boundary of $K$.
\begin{lemen}\label{lem1.4}
	\emph{\cite{Assad}}
	If $K$ is a closed subset of the complete and convex metric space $M$ and if $u\in K$, $v\notin K$, then there exists a point $z\in \partial K$, such that \[\varrho(u,v)=\varrho(u,z)+\varrho(z,v),\]
	where $\partial K$ denotes the boundary of $K$.
\end{lemen}
Assad and Kirk \cite{Assad} proved the following fixed point theorem.
\bigskip\noindent
\begin{theoen}\emph{\cite{Assad}}\label{thm1.5}
	Let $(M, \varrho)$ be a complete and metrically convex metric space, $K$ be a non-empty closed subset of $M$, $T : K \to CB(M)$ be a mapping such that, for all $u,v\in K$,
	\[\rho(Tu, Tv)\leq k \varrho(u, v),\]  
	for some $k\in (0, 1)$. If $Tu\subseteq K$ for each $u\in \partial K$, then $T$ has a fixed point in $K$.
\end{theoen}
In 2014, Jleli and Sameti \cite{Jleli} gave a new generalization of Banach contraction mapping theorem in the setting of Banciari metric spaces as follows:
\bigskip\noindent
\begin{theoen}\emph{\cite{Jleli}}\label{thm1.6}
	Let $(M, \varrho)$ be a complete generalized metric space and $T:M\to M$ be a mapping. Suppose that there exists $\vartheta \in \varTheta$ and $k\in (0,1)$ such that $u,v\in M$, \[\varrho(Tu,Tv)\neq 0 \Longrightarrow \vartheta(\varrho(Tu,Tv))\leq [\vartheta(\varrho(u,v))]^{k}\]
	Then $T$ has a unique fixed point.
\end{theoen}
Recently, Altun and Minak \cite{Altun} obtained a new approach to Theorem \ref{thm1.6} and a new generalization of it, by using $\vartheta$-contraction as follows: 
\begin{theoen}\emph{\cite{Altun}}\label{thm1.7}
	Let $(M,\varrho)$ be a complete and metrically convex metric space, $K$ be a nonempty closed subset of $M$, $T:K\to CB(M)$ be a mapping such that for all $u,v\in K$ with
	$\mathcal{H}(Tu,Tv)>0$,  \[\vartheta(\mathcal{H}(Tu,Tv))\leq [\vartheta(\varrho(u,v))]^{k}\] for some $k\in (0,1)$ and $\vartheta \in \varTheta$. If $Tu\subseteq K$ for each $u\in \partial K$, then $T$ has a fixed point in $K$. 
\end{theoen}
Suppose we want to use a different $\vartheta$-contraction in the above theorem. Hussain \emph{et al.} \cite{Hussain} introduced a new concept that we can apply as in the proof of the above theorem and obtain a new result.
\begin{dfnen}\emph{\cite{Hussain}}
	Let $(M, \varrho)$ be a metric space and let $T:M\to M$ be a mapping. $T$ is said to be a JS-contraction whenever there are a function $\vartheta \in \varTheta$ and positive real numbers $\tau_{1}, \tau_{2}, \tau_{3}, \tau_{4}$ with $0\leq \tau_{1}+\tau_{2}+\tau_{3}+2\tau_{4}<1$ such that
	\begin{equation}
		\begin{split}
			\vartheta(\varrho(Tu,Tv))&\leq [\vartheta(\varrho(u,v))]^{\tau_{1}}[\vartheta(\varrho(u,Tu))]^{\tau_{2}}[\vartheta(\varrho(v,Tv))]^{\tau_{3}}\\
			&\hspace*{0.5cm} \times [\vartheta(\varrho(u,Tv)+\varrho(v,Tu))]^{\tau_{4}}
		\end{split}
	\end{equation} for all $u,v\in M$.
\end{dfnen}



\section{JS-contraction fixed point theorem}
\setcounter{equation}{0}

The following is a definition of a generalized multivalued JS-contraction mapping.

\begin{dfnen}
	Let $(M,\varrho)$ be a metric space and $K$ be a nonempty closed subset of  $M$. Let $T$ be a mapping of $K$ into $CB(M)$. Then $T$ is said to be a generalized JS-contraction mapping whenever there are a function $\vartheta \in \varTheta$ and nonnegative real numbers $\tau_{1}, \tau_{2}, \tau_{3}, \tau_{4}$ with $0\leq \tau_{1}+\tau_{2}+\tau_{3}+2\tau_{4}<1$ such that
	\begin{equation}\label{eq2.1}
		\begin{split}
			\vartheta(\mathcal{H}(Tu,Tv))&\leq [\vartheta(\varrho(u,v))]^{\tau_{1}}[\vartheta(\varrho(u,Tu))]^{\tau_{2}}[\vartheta(\varrho(v,Tv))]^{\tau_{3}}\\
			&\hspace*{0.5cm} \times [\vartheta(\varrho(u,Tv)+\varrho(v,Tu))]^{\tau_{4}}.
		\end{split}
	\end{equation}  for all $u,v\in K$.
\end{dfnen}
We now present an extended version of Theorem \ref{thm1.7}. 
\bigskip\noindent
\begin{theoen}\label{thm2.2} Let $(M,\varrho)$ be a complete and metrically convex metric space, $K$ be a nonempty closed subset of $M$. Let $T:K\to CB(M)$ be a generalized multivalued JS-contraction mapping. If for any $u\in \partial K$, $Tu\subseteq K$ and \[\cfrac{(1+\tau_{1}+\tau_{2}+\tau_{4})(\tau_{1}+\tau_{2}+\tau_{4})}{(1-\tau_{3}-\tau_{4})^{2}}<1,\] then, there is a $z\in K$ such that $z\in T(z)$. 
\end{theoen}

\begin{proof}
	We construct two sequences $\{u_n\}$ and $\{v_{n}\}$ in $K$ in the following way:
	Let $u_0\in K$ and $v_{1}\in Tu_{0}$. If $v_{1}\in K$, let $u_{1}=v_{1}$. If $v_{1}\notin K$, then from Lemma \ref{lem1.4}, there exists $u_{1}\in\partial K$ such that
	\[\varrho(u_{0},u_{1})+\varrho(u_{1},v_{1})=\varrho(u_{0},v_{1}).\]
	Thus, $u_1\in K$. Now, we claim that $\varrho(v_{1},Tu_{1})\geq 0$. Suppose $\varrho(v_{1},Tu_{1})=0$.
	If $v_{1}\in K$, then $u_{1}$ is a fixed point of $T$, which is a contradiction. If $v_{1}\notin K$, then $u_{1}\in \partial K$ and so $Tu_{1}\subseteq K$. Therefore, $v_{1}\notin Tu_{1}$, which is a contradiction. Thus, $\varrho(v_{1},Tu_{1})\geq 0$. Now, since $\varrho(v_{1},Tu_{1})\leq \mathcal{H}(Tu_{0},Tu_{1})$, then we have
	\begin{equation}\label{eq2.2}
		\begin{split}
			\vartheta(\varrho(v_{1},Tu_{1}))&\leq \vartheta(\mathcal{H}(Tu_{0},Tu_{1}))\\&\leq [\vartheta(\varrho(u_{0},u_{1}))]^{\tau_{1}}[\vartheta(\varrho(u_{0},Tu_{0}))]^{\tau_{2}}[\vartheta(\varrho(u_{1},Tu_{1}))]^{\tau_{3}}\\
			&\ \times [\vartheta(\varrho(u_{0},Tu_{1})+\varrho(u_{1},Tu_{0}))]^{\tau_{4}}.
		\end{split}
	\end{equation}
	On the other hand, from $\vartheta _{5}$ we get
	\[\vartheta(\varrho(v_{1},Tu_{1}))=\vartheta(\inf \{\varrho(v_{1},m):m\in Tu_{1}\})=\inf \{\vartheta(\varrho(v_{1},m)):m\in Tu_{1}\}\]
	and so from condition (\ref{eq2.2}) we get
	\begin{align}
		\begin{split}
			\inf \{\vartheta(\varrho(v_{1},m)):m\in Tu_{1}\}&\leq [\vartheta(\varrho(u_{0},u_{1}))]^{\tau_{1}}[\vartheta(\varrho(u_{0},Tu_{0}))]^{\tau_{2}}[\vartheta(\varrho(u_{1},Tu_{1}))]^{\tau_{3}}\\
			&\ \times [\vartheta(\varrho(u_{0},Tu_{1})+\varrho(u_{1},Tu_{0}))]^{\tau_{4}}.
		\end{split}
	\end{align}
	Thus, there exists $v_{2}\in Tu_{1}$ such that 
	\begin{align}
		\begin{split}
			\vartheta(\varrho(v_{1},v_{2}))&\leq [\vartheta(\varrho(u_{0},u_{1}))]^{\gamma_{1}}[\vartheta(\varrho(u_{0},Tu_{0}))]^{\gamma_{2}}[\vartheta(\varrho(u_{1},Tu_{1}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{0},Tu_{1})+\varrho(u_{1},Tu_{0}))]^{\gamma_{4}},
		\end{split}
	\end{align}
	where  $0\leq \tau_{1}+\tau_{2}+\tau_{3}+2\tau_{4}<\gamma_{1}+\gamma_{2}+\gamma_{3}+2 \gamma_{4}<1$.\\
	If $v_{2}\in K$ let $u_{2}=v_{2}$. If $v_{2}\notin K$, select a point $u_{2}\in \partial K$ such that \[\varrho(u_{1},u_{2})+\varrho(u_{2},v_{2})=\varrho(u_{1},v_{2}).\]
	Thus, $u_{2}\in K$. We can show that $\varrho(v_{2},Tu_{2})>0$. As above, we can find a point $v_{3}\in Tu_{2}$ such that
	\begin{align}
		\begin{split}
			\vartheta(\varrho(v_{2},v_{3}))&\leq [\vartheta(\varrho(u_{1},u_{2}))]^{\gamma_{1}}[\vartheta(\varrho(u_{1},Tu_{1}))]^{\gamma_{2}}[\vartheta(\varrho(u_{2},Tu_{2}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{1},Tu_{2})+\varrho(u_{2},Tu_{1}))]^{\gamma_{4}}.
		\end{split}
	\end{align}
	Continuing the arguments, two sequences $\{u_{n}\}$ and $\{v_{n}\}$ are obtained such that for $n\in N$
	
	(i) $v_{n+1}\in Tu_{n}$
	
	(ii) $
	\vartheta(\varrho(v_{n},v_{n+1}))\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},Tu_{n-1}))]^{\gamma_{2}}[\vartheta(\varrho(u_{n},Tu_{n}))]^{\gamma_{3}}\\
	\hspace*{4cm}\times [\vartheta(\varrho(u_{n-1},Tu_{n})+\varrho(u_{n},Tu_{n-1}))]^{\gamma_{4}},
	$
	
	where $v_{n+1}=u_{n+1}$ if $v_{n+1}\in K$ or
	\begin{equation}\label{eq2.6}
		\varrho(u_{n},u_{n+1})+\varrho(u_{n+1},v_{n+1})=\varrho(u_{n},v_{n+1})
	\end{equation}
	if $v_{n+1}\notin K$ and $u_{n+1}\in \partial K$.
	
	Now, we set\\
	$P=\{u_{\xi}\in \{u_{n}\}:u_{\xi}= v_{\xi}, \xi\in \mathbb{N}\}$ and $Q=\{u_{\xi}\in \{u_{n}\}:u_{\xi}\neq v_{\xi}, \xi\in \mathbb{N}\}$.\\
	Observe that if $u_{\xi}\in Q$ for some $\xi$, then $u_{\xi+1}\in P$.\\
	Here, the intention is to estimate the distance $\varrho(u_{n},u_{n+1})$ for $n\geq 2$. Note that $\varrho(u_{n},u_{n+1})>0$, otherwise, $T$ has a fixed point. For this, three cases have to be considered:
	
	\textbf{Case 1.} If $u_{n}\in P$ and $u_{n+1}\in P$, then, we get 
	\begin{align*}
		\begin{split}
			\vartheta(\varrho(u_{n},u_{n+1}))&=\vartheta(\varrho(v_{n},v_{n+1}))\\
			&\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},Tu_{n-1}))]^{\gamma_{2}}[\vartheta(\varrho(u_{n},Tu_{n}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{n-1},Tu_{n})+\varrho(u_{n},Tu_{n-1}))]^{\gamma_{4}}\\
			&=[\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{2}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{n-1},u_{n+1})+\varrho(u_{n},u_{n}))]^{\gamma_{4}}\\
			&=[\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}+\gamma_{2}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{n-1},u_{n})+\varrho(u_{n},u_{n+1}))]^{\gamma_{4}}\\
			&\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}+\gamma_{2}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{n-1},u_{n})]^{\gamma_{4}}[\varrho(u_{n},u_{n+1}))]^{\gamma_{4}}\\ 
			&=[\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}+\gamma_{2}+\gamma_{4}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}+\gamma_{4}}
		\end{split}
	\end{align*}
	It follows that \[\vartheta(\varrho(u_{n},u_{n+1}))\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\frac{\gamma_{1}+\gamma_{2}+\gamma_{4}}{1-\gamma_{3}-\gamma_{4}}}\]
	\textbf{Case 2.} If $u_{n}\in P$ and $u_{n+1}\in Q$, then, from condition (\ref{eq2.6}), we get 
	\begin{align*}
		\vartheta(\varrho(u_{n},u_{n+1}))&\leq \vartheta(\varrho(u_{n},u_{n+1})+\varrho(u_{n+1},v_{n+1}))=\vartheta(\varrho(v_{n},v_{n+1}))\\
		&\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\frac{\gamma_{1}+\gamma_{2}+\gamma_{4}}{1-\gamma_{3}-\gamma_{4}}}
	\end{align*}
	\textbf{Case 3.} If $u_{n}\in Q$ and $u_{n+1}\in P$, then, since
	\begin{align*}
		\begin{split}
			\vartheta(\varrho(v_{n},v_{n+1}))&\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},Tu_{n-1}))]^{\gamma_{2}}[\vartheta(\varrho(u_{n},Tu_{n}))]^{\gamma_{3}}\\
			&\ \times [\vartheta(\varrho(u_{n-1},Tu_{n})+\varrho(u_{n},Tu_{n-1}))]^{\gamma_{4}},
		\end{split}
	\end{align*}
	$\Rightarrow$
	\begin{align*}
		\varrho(v_{n},v_{n+1})&<(\varrho(u_{n-1},u_{n}))(\varrho(u_{n-1},Tu_{n-1}))(\varrho(u_{n},Tu_{n}))\\
		&\ \times (\varrho(u_{n-1},Tu_{n})+\varrho(u_{n},Tu_{n-1}))
	\end{align*}
	In our case, if we simplify we get
	\begin{align*}
		\vartheta(\varrho(v_{n},v_{n+1}))&\leq [\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{2}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}}\\
		&\ \times [\vartheta(\varrho(u_{n-1},u_{n+1})+\varrho(u_{n},v_{n}))]^{\gamma_{4}}\\
		&\leq [\vartheta(\varrho(u_{n-1},u_{n})+\varrho(u_{n},v_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{2}}\times\\
		&\ \hspace*{0.4cm}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}} [\vartheta(\varrho(u_{n-1},u_{n})+\varrho(u_{n},u_{n+1})+\varrho(u_{n},v_{n}))]^{\gamma_{4}}\\
		&\leq [\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{1}}[\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{2}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}}\\
		&\ \times [\vartheta(\varrho(u_{n-1},v_{n})+\varrho(u_{n},u_{n+1}))]^{\gamma_{4}}\\
		&\leq [\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{1}+\gamma_{2}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}}\\
		&\ \times [\vartheta(\varrho(u_{n-1},v_{n})]^{\gamma_{4}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{4}}\\
		&= [\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{1}+\gamma_{2}+\gamma_{4}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}+\gamma_{4}}
	\end{align*}
	It follows that
	\begin{align*}
		\vartheta(\varrho(u_{n},u_{n+1}))&\leq \vartheta(\varrho(u_{n},v_{n})+\varrho(v_{n},u_{n+1}))\\
		&< \vartheta(\varrho(u_{n-1},u_{n})+\varrho(u_{n},v_{n})+\varrho(v_{n},v_{n+1}))\\
		&=\vartheta(\varrho(u_{n-1},v_{n})+\varrho(v_{n},v_{n+1}))\\
		&\leq \vartheta(\varrho(u_{n-1},v_{n}))\vartheta(\varrho(v_{n},v_{n+1}))\\
		&\leq \vartheta(\varrho(u_{n-1},v_{n}))[\vartheta(\varrho(u_{n-1},v_{n}))]^{\gamma_{1}+\gamma_{2}+\gamma_{4}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}+\gamma_{4}}\\
		&=[\vartheta(\varrho(u_{n-1},v_{n}))]^{1+\gamma_{1}+\gamma_{2}+\gamma_{4}}[\vartheta(\varrho(u_{n},u_{n+1}))]^{\gamma_{3}+\gamma_{4}}
	\end{align*}
	Hence
	\begin{align*}
		[\vartheta(\varrho(u_{n},u_{n+1}))]^{1-\gamma_{3}-\gamma_{4}}&\leq[\vartheta(\varrho(u_{n-1},v_{n}))]^{1+\gamma_{1}+\gamma_{2}+\gamma_{4}}
	\end{align*}
	by Case 2, since $u_{n}\in Q$ implies $u_{n-1}\in P$ we have 
	\[\vartheta(\varrho(u_{n-1},v_{n}))\leq [\vartheta(\varrho(u_{n-2},u_{n-1}))]^{\frac{\gamma_{1}+\gamma_{2}+\gamma_{4}}{1-\gamma_{3}-\gamma_{4}}}\]
	Therefore,
	\begin{align*}
		\vartheta(\varrho(u_{n},u_{n+1}))&\leq[\vartheta(\varrho(u_{n-2},u_{n-1}))]^{\frac{(1+\gamma_{1}+\gamma_{2}+\gamma_{4})(\gamma_{1}+\gamma_{2}+\gamma_{4})}{(1-\gamma_{3}-\gamma_{4})^{2}}}
	\end{align*}
	The case that $u_{n}\in Q$ and $u_{n+1}\in Q$ does not occur.
	
	Since $\dfrac{\gamma_{1}+\gamma_{2}+\gamma_{4}}{1-\gamma_{3}-\gamma_{4}}\leq \dfrac{(1+\gamma_{1}+\gamma_{2}+\gamma_{4})(\gamma_{1}+\gamma_{2}+\gamma_{4})}{(1-\gamma_{3}-\gamma_{4})^{2}}$, for $n\geq 2$ we have
	
	\begin{equation}
		\vartheta(\varrho(u_{n},u_{n+1}))\leq 
		\begin{cases}
			[\vartheta(\varrho(u_{n-1},u_{n}))]^{\gamma}\\
			[\vartheta(\varrho(u_{n-2},u_{n-1}))]^{\gamma}.
		\end{cases}
	\end{equation}
	Now we claim that 
	\begin{equation}\label{eq2.8}
		\vartheta(\varrho(u_{n},u_{n+1}))\leq \delta ^{\big(\gamma^{\frac{n-1}{2}}\big)}
	\end{equation}
	for all $n\in \mathbb{N}$, where 
	\[\delta = \max \{\vartheta(\varrho(u_{0},u_{1})),\vartheta(\varrho(u_{1},u_{2}))\}.\]
	Using Equation (\ref{eq2.8}) we obtain
	\begin{equation}
		\lim \limits_{n\to \infty} \vartheta(\varrho(u_{n},u_{n+1}))=1.
	\end{equation}
	From $(\vartheta _{2})$, $\lim \limits_{n\to \infty} \varrho(u_{n},u_{n+1})= 0$ and so from $(\vartheta _{3})$ there exists $r\in (0,1)$ and $l\in (0,\infty]$ such that \[\lim \limits_{n\to \infty} \dfrac{\vartheta(\varrho(u_{n},u_{n+1}))-1}{[\varrho(u_{n},u_{n+1})]^{r}}=l.\]
	Suppose that $l<\infty$. In this case, let $\varPsi=\frac{l}{2}>0$. Recall from the definition of the limit, there exists $n_{0}\in \mathbb{N}$ such that, for all $n\geq n_{0}$,
	$$\bigg \lvert \dfrac{\vartheta(\varrho(u_{n},u_{n+1}))-1}{[\varrho(u_{n},u_{n+1})]^{r}}-l\bigg \rvert \leq \varPsi.$$
	This implies that, for all $n\geq n_{0}$,
	\[\dfrac{\vartheta(\varrho(u_{n},u_{n+1}))-1}{[\varrho(u_{n},u_{n+1})]^{r}}\geq l-\varPsi=\varPsi.\]
	Then, for all $n\geq n_{0}$,
	\[n[\varrho(u_{n},u_{n+1})]^{r}\leq \varPhi n[\vartheta(\varrho(u_{n},u_{n+1}))-1],\]
	where $\varPhi = 1/\vartheta.$\\
	Thus, in all cases, there exist $\varPhi>0$ and $n_{0}\in \mathbb{N}$ such that, for all $n\geq n_{0}$,
	\[n[\varrho(u_{n},u_{n+1})]^{r}\leq \varPhi n[\vartheta(\varrho(u_{n},u_{n+1}))-1].\]
	Using Equation (\ref{eq2.8}), we obtain, for all $n\geq n_{0}$,
	\[n[\varrho(u_{n},u_{n+1})]^{r}\leq \varPhi n[\delta ^{\big(\gamma^{\frac{n-1}{2}}\big)}-1].\] 
	Letting $n\to \infty$ in the above inequality, we obtain 
	\[\lim \limits_{n\to \infty}n[\varrho(u_{n},u_{n+1})]^{r}=0.\]
	Thus, there exists $n_{1}\in \mathbb{N}$ such that $n[\varrho(u_{n},u_{n+1})]^{r}\leq 1$ for all $n\geq n_{1}$. So, we have, for all $n\geq n_{0}$
	\begin{equation}\label{eq2.10}
		\varrho(u_{n},u_{n+1})\leq \dfrac{1}{n^{\frac{1}{r}}}.
	\end{equation}
	In order to demonstrate that $\{x_{n}\}$ is a Cauchy sequence consider $m,n\in \mathbb{N}$ such that $m>n\geq n_{1}$. Now, applying the metric triangle inequality and from condition (\ref{eq2.10}), we have
	\begin{align*}
		\varrho(u_{n},u_{m})&\leq \varrho(u_{n},u_{n+1})+\varrho(u_{n+1},u_{n+2})+\cdots+\varrho(u_{m-1},u_{m})\\
		&= \sum_{\xi=n}^{m-1} \varrho(u_{\xi},u_{\xi+1})\leq \sum_{\xi=n}^{\infty} \varrho(u_{\xi},u_{\xi+1})\leq \sum_{\xi=n}^{\infty} \dfrac{1}{\xi^{1/r}}.
	\end{align*}
	Since the series $\sum_{\xi=1}^{\infty} \dfrac{1}{\xi^{1/r}}$ converges, then passing to limit $n,m\to \infty$, we get $\varrho(u_{n},u_{m})\to 0$.
	
	It is an obvious implication that the sequence $\{u_{n}\}$ is a Cauchy sequence in $K$. Since $K$ is closed, the sequence $\{u_{n}\}$ converges to some point $z\in K$. By our choice of $\{u_{n}\}$, there exists a subsequence  $\{u_{n_{k}}\}$ of $\{u_{n}\}$ such that  $\{u_{n_{k}}\}\in P$ that is, $\{u_{n_{k}}\}=\{v_{n_{k}}\}$, $k\in \mathbb{N}$. Note that $\{u_{n_{k}}\}\in \{u_{n_{k}-1}\}$ for $k\in \mathbb{N}$ and $\{u_{n_{k}}\}\to z$ as $k\to \infty$. Also note that from condition (\ref{eq2.1}) and $(\vartheta_{1})$ we get 
	\[
	\mathcal{H}(Tu,Tv)\leq (\varrho(u,v))(\varrho(u,Tu))(\varrho(v,Tv))(\varrho(v,Tu)+\varrho(u,Tv))
	\]
	for all $u,v\in K$ and so, we have
	\begin{align*}
		\varrho(u_{n_{k}},Tz)&\leq \mathcal{H}(Tu_{n_{k}-1},Tz)\\&\leq \varrho(u_{n_{k}-1},z)\varrho(u_{n_{k}-1},Tu_{n_{k}-1})\varrho(z,Tz)\\
		&\ \cdot \{\varrho(z,Tu_{n_{k}-1})+\varrho(u_{n_{k}-1},Tz)\}
	\end{align*}
	which on letting $k\to \infty$ implies that $\varrho(z,Tz)=0$, which is a contradiction.  Therefore, $T$ has a fixed point $z\in K$.
\end{proof}

\textbf{Remark 2.1} If $\tau_{2}=\tau_{3}=\tau_{4}=0$ and $\tau_{1}=\tau$ in Theorem \ref{thm2.2} we obtain Theorem \ref{thm1.7} of Altun \cite{Altun}.

For particular function $\vartheta$ selections, some significant results are obtained. First, by setting $\vartheta(\mu)=e^{\sqrt{\mu}}$ in Theorem \ref{thm2.2}, the following corollary is obtained:
\begin{coren}
	Let $K$ be a nonempty closed subset of a complete and metrically convex metric space $M$. Let $T:K\to CB(M)$ be a mapping such that the following condition holds:
	\begin{equation*}
		\begin{split}
			\sqrt{\mathcal{H}(Tu,Tv)}&\leq \tau_{1}\sqrt{\varrho(u,v)}+\tau_{2}\sqrt{\varrho(u,Tu)}+\tau_{3}\sqrt{\varrho(v,Tv)}\\
			&\ \hspace*{1cm} + \tau_{4}\sqrt{\varrho(u,Tv)+\varrho(v,Tu)}
		\end{split}
	\end{equation*} 
	for all $u,v\in K$,  $\vartheta \in \varTheta$ and $\tau_{1}, \tau_{2}, \tau_{3}, \tau_{4}\geq 0$ with $0\leq \tau_{1}+\tau_{2}+\tau_{3}+2\tau_{4}<1$. Then $T$ has a unique fixed point.		
\end{coren}
And, by putting $\vartheta(\mu)=e^{\sqrt[n]{\mu}}$ in Theorem \ref{thm2.2}, the following corollary is obtained:
\begin{coren}
	Let $K$ be a nonempty closed subset of a complete and metrically convex metric space $M$. Let $T:K\to CB(M)$ be a mapping such that the following condition holds:
	\begin{equation*}
		\begin{split}
			\sqrt[n]{\mathcal{H}(Tu,Tv)}&\leq \tau_{1}\sqrt[n]{\varrho(u,v)}+\tau_{2}\sqrt[n]{\varrho(u,Tu)}+\tau_{3}\sqrt[n]{\varrho(v,Tv)}\\
			&\ \hspace*{1cm} + \tau_{4}\sqrt[n]{\varrho(u,Tv)+\varrho(v,Tu)}
		\end{split}
	\end{equation*} 
	for all $u,v\in K$,  $\vartheta \in \varTheta$ and $\tau_{1}, \tau_{2}, \tau_{3}, \tau_{4}\geq 0$ with $0\leq \tau_{1}+\tau_{2}+\tau_{3}+2\tau_{4}<1$. Then $T$ has a unique fixed point.		
\end{coren}

\section{Application to Nonlinear Integral Equations}
Nonlinear integral equations can be solved using a variety of numerical approaches.
The integral equation is usually transformed into a system of nonlinear algebraic equations. Solving these systems is difficult, or the solution may be impossible to find.
Therefore, in this section, we look at how the fixed point approach may be used to solve Volterra-Hammerstein integral equations. This approach does not result in a system of nonlinear algebraic equations.

Now, consider the nonlinear integral equation below:
\begin{equation}\label{eq3.1}
	u(t)=g(t)+\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,u(\tau))d\tau,
\end{equation}
where $t,\tau \in [a,b]$, $a,b\in \mathbb{R}$, $u\in C[a,b]$, $g:[a,b]\to \mathbb{R}$, $\mathcal{H}\in C[a,b]\times \mathbb{R} \to \mathbb{R}$ and $k\in C^{2}[a,b]^{2}$ such that $k(t,\tau)>0$ are given functions.

Maleknejad \cite{Maleknejad} established some conditions which ensure the uniqueness of the solution and how the fixed point method approximates this solution.

Referring from Maleknejad \cite{Maleknejad} we are going to establish the following fixed point theorem.
\bigskip\noindent
\begin{theoen}
	Let $M=C[a,b]$ be a metric space endowed with the metric $$\varrho(u,v)=\sup_{t\in [a,b]} |u(t)-v(t)|.$$ Define the mapping $T:K\to CB(M)$ by $T(u)(t)=g(t)+\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,u(\tau))d\tau.$ Let $u,v\in K$ and $t\in [a,b]$. Assume that $g\in C[a,b]$, $k\in C^{2}[a,b]^{2}$, i.e. there exists a constant $M>0$ where \[\bigg(\int_{a}^{b}k^{2}(t,\tau)d\tau\bigg)^{\frac{1}{2}}\leq M<\infty,\] and $\mathcal{H}:[a,b]\times \mathbb{R}\to \mathbb{R}$ is continuous and there is $\vartheta \in \varTheta$ so that $\vartheta (\sup f(t))=\sup \vartheta (f(t))$ for arbitrary function $f$ with 
	\[\vartheta (\int_{a}^{b}|\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau))|d\tau)\leq \int_{a}^{b}\vartheta(|\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau))|)d\tau\]
	there is $\tau_{i}\in (0,1)$ where $i=1,2,3,4$ such that 
	\begin{align*}
		\vartheta(|\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau))|)\\
		&\hspace*{-5cm}\leq \bigg\{[\vartheta(|u(t)-v(t)|)]^{\tau_{1}}[\vartheta(|u(t)-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,u(\tau))d\tau|)]^{\tau_{2}}[\vartheta(|v(t)\\
		&\hspace*{-5cm}-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,v(\tau))d\tau|)]^{\tau_{3}}
		[\vartheta(|u(t)-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,v(\tau))d\tau|+|v(t) \\&-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,u(\tau))d\tau|)]^{\tau_{4}}\bigg\}\bigg /M(b-a)
	\end{align*}
	Then Equation (\ref{eq3.1}) has a unique solution.
\end{theoen}
\begin{proof}
	We begin our proof by deriving the following relation where Cauchy-Schwartz inequality is used:
	\begin{align*}
		|Tu(t)-Tv(t)|\\
		&\hspace*{-3cm}=\bigg \lvert \int_{a}^{b}k(t,\tau)(\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau)))d\tau\bigg \rvert\\
		&\hspace*{-3cm}\leq \int_{a}^{b}\lvert k(t,\tau) \rvert \lvert \mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau)))d\tau \rvert\\
		&\hspace*{-3cm}\leq \bigg(\int_{a}^{b}k^{2}(t,\tau)d\tau\bigg)^{\frac{1}{2}}\bigg(\int_{a}^{b}\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau)))d\tau\bigg)^{\frac{1}{2}}\\
		&\hspace*{-3cm}\leq M \bigg(\int_{a}^{b}\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau)))d\tau\bigg)^{\frac{1}{2}}\\
		&\hspace*{-3cm}\leq M \int_{a}^{b}\lvert \mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau)))\rvert d\tau
	\end{align*}
	Now, we have 
	\begin{align*}
		\vartheta(|Tu(t)-Tv(t)|)\\
		&\hspace*{-3cm}=\vartheta (M\int_{a}^{b}|\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau))|d\tau)\\
		&\hspace*{-3cm}\leq M \int_{a}^{b}\vartheta(|\mathcal{H}(\tau,u(\tau))-\mathcal{H}(\tau,v(\tau))|)d\tau\\
		&\hspace*{-3cm}\leq \bigg\{ M [\vartheta(|u(t)-v(t)|)]^{\tau_{1}}[\vartheta(|u(t)-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,u(\tau))d\tau|)]^{\tau_{2}}\\
		&\hspace*{-2.5cm}
		[\vartheta(|v(t)-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,v(\tau))d\tau|)]^{\tau_{3}} [\vartheta(|u(t)-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,v(\tau))d\tau|\\
		&\hspace*{-2.5cm}
		+|v(t)-\int_{a}^{b}k(t,\tau)\mathcal{H}(\tau,u(\tau))d\tau|)]^{\tau_{4}}\bigg\}\bigg /M(b-a)\\
		&\hspace*{-3cm}\leq \dfrac{1}{b-a}\int_{a}^{b} [\vartheta(\varrho(u,v))]^{\tau_{1}}[\vartheta(\varrho(u,Tu))]^{\tau_{2}}[\vartheta(\varrho(v,Tv))]^{\tau_{3}}[\vartheta(\varrho(u,Tv)+\varrho(v,Tu))]^{\tau_{4}}\\
		&\hspace*{-3cm}=[\vartheta(\varrho(u,v))]^{\tau_{1}}[\vartheta(\varrho(u,Tu))]^{\tau_{2}}[\vartheta(\varrho(v,Tv))]^{\tau_{3}}[\vartheta(\varrho(u,Tv)+\varrho(v,Tu))]^{\tau_{4}}
	\end{align*}
	Thus, all the conditions of Theorem \ref{thm2.2} are satisfied. Hence the integral Equation (\ref{eq3.1}) has a solution.
\end{proof}

\section{Conclusion}
The main contribution of this study is Definition 2.1 and Theorem 2.2. This theorem is proved for multivalued non-self mappings in complete and metrically convex space. This theorem generalizes the fixed point theorem due to Altun and Minak \cite{Altun} by using $\vartheta$-contraction due to Jleli and Sameti \cite{Jleli}. To validate the results proved here, we provide our main result an appropriate application. 

{\bf{Declarations:}}

{\bf{Availability of data and material:}}
This clause is not applicable to this paper.

{\bf{Competing interests:}}
The authors declare that they have no competing interests.

{\bf{Funding:}}
Authors declare that there is no funding available for this article.

{\bf{Author's contributions:}}
All authors contributed equally and significantly in writing this article. All authors read and approved the final manuscript.

{\bf{Acknowledgment:}} The authors are thankful to the learned reviewer for his valuable comments.




\end{document}
